Author's response to reviews

Title: How do older patients and their GPs evaluate shared decision-making in health care?

Authors:

Danica Rotar-Pavlic (danica.rotar@guest.arnes.si)
Igor Svab (igor.svab@mf.uni-lj.si)
Raymond Wetzels (r.wetzels@kwazo.umcn.nl)

Version: 5 Date: 25 April 2008

Author's response to reviews: see over
Dear editor,

The editorial production team has requested some revisions to our manuscript. On behalf of the authors I am sending the revision of the article **How do older patients and their GPs evaluate shared decision-making in healthcare?**

The figures are cropped as closely as possible to minimise white space around the image. All additional files are mentioned in the text, and referred. A section titled "Additional files" is added at the end of the manuscript (after the tables).

Yours sincerely,

Danica Rotar Pavlič